Author's response to reviews

Title: Recognizing Acute Delirium As part of your Routine [RADAR]: A validation study

Authors:

Philippe Voyer (philippe.voyer@fsi.ulaval.ca)
Nathalie Champoux (nathalie.champoux@umontreal.ca)
Johanne Desrosiers (johanne.desrosiers@usherbrooke.ca)
Philippe Landreville (philippe.landreville@psy.ulaval.ca)
Jane McCusker (jane.mccusker@mcgill.ca)
Johanne Monette (johanne.monette@mcgill.ca)
Maryse Savoie (maryse.savoie@vac-acc.gc.ca)
Sylvie Richard (sylvie.richard@fsi.ulaval.ca)
Pierre-Hugues Carmichael (pierre-hugues.carmichael.cha@ssss.gouv.qc.ca)

Version: 3 Date: 11 November 2014

Author's response to reviews:

As requested, we have adjusted the format of the manuscript to follow the BMC Nursing Guidelines (line and page numbers were added).